Author’s response to reviews

Title: Immunogenomic Pathways Associated with Cytotoxic Lymphocyte Infiltration and Survival in Colorectal Cancer.

Authors:
Yuanyuan Shen (yshpf@mail.missouri.edu)
Yue Guan (guanyue@health.missouri.edu)
Justin J. Hummel (jjh42v@mail.missouri.edu)
Chi-Ren Shyu (shyuc@missouri.edu)
Jonathan B. Mitchem (mitchemj@health.missouri.edu)

Version: 3 Date: 26 Dec 2019

Author’s response to reviews:

Dear Professor Kutuk-

Thank you for your remarks and the further consideration of our work. Please see the comments below.

Editor Comments:

1 - Please move the Funding information from the Acknowledgements to the Funding section.

This has been corrected.

2 - If you wish to acknowledge someone by name in the Acknowledgements, please ensure you have obtained permission from them to do so.

This has been completed.

3 - At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.

This has been completed to the best of our knowledge. Figures have been removed from the manuscript and uploaded separately as PDFs.
Thank you again for your consideration and help in improving our work.

Sincerely,

Jonathan B. Mitchem